Author's response to reviews

Title: The efficacy of risk scores for predicting abdominal wound dehiscence: a case-controlled validation study.

Authors:

Jakub Kenig (jkenig@cm-uj.krakow.pl)
Piotr Richter (jkenig@cm-uj.krakow.pl)
Anna Lasek (aniad303@gmail.com)
Katarzyna Zbierska (alistra16@gmail.com)
Sabina Zurawska (s.zurawska@gmail.com)

Version: 7 Date: 16 July 2014

Author's response to reviews: see over
We would like to submit a revised version of the attached manuscript.

First of all, we would like to thank you the reviewer for the valuable comments. We analyzed very carefully each point and we hope that we were able to completely and satisfactorily answer each of them. All of the comments were addressed and we made appropriate changes in the text, highlighting it.

Sincerely,

Jakub Kenig

Comments of reviewer Gabrielle van Ramshorst,
Major compulsory revision
Be more specific in method section about how data was extracted from which source and explain in more detail 'variables/factors of interest'. Added
Minor essential revisions:
Please check entire manuscript using spelling control. Corrected

Editorial comments:
The retrospective access to the database was approved by the Department’s review board. We included such a information in our manuscript.